Page: 1 Total Pages: 10 ini:

EMBARGOED FOR RELEASE UNTIL 5:00 PM ET ON MONDAY 14 JULY 2025

Annals of Internal Medicine

ORIGINAL RESEARCH

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

A Nationwide Cohort Study

Niklas Worm Andersson, MD, PhD; Ingrid Bech Svalgaard, MSc; Stine Skovbo Hoffmann, MD, PhD; and Anders Hviid, MSc, DMSc

Background: Aluminum is used as an adjuvant in nonlive vaccines administered in early childhood. Concerns persist about potential associations between vaccination with aluminum-adsorbed vaccines and increased risk for chronic autoimmunity, atopy or allergy, and neurodevelopmental disorders. Large-scale safety data remain limited.

Objective: To assess the association between cumulative aluminum exposure from early childhood vaccination and risk for autoimmune, atopic or allergic, and neurodevelopmental disorders.

Design: A cohort study linking nationwide registry data on childhood vaccinations, outcome diagnoses, and potential confounders, leveraging the variations in the aluminum content of childhood vaccines over time.

Setting: Denmark, 1997 to 2020.

Participants: 1224176 children born in Denmark between 1997 and 2018 who were alive and residing in the country at age 2 years.

Intervention: Cumulative aluminum amount received (per 1-mg increase) through vaccination during the first 2 years of life.

Measurements: Incident events of 50 chronic disorders, including autoimmune (dermatologic, endocrinologic, hematologic, gastrointestinal, and rheumatic), atopic or allergic (asthma, atopic dermatitis, rhinoconjunctivitis, and allergy), and neurodevelopmental (autism spectrum disorder and attention deficithyperactivity disorder).

Results: Cumulative aluminum exposure from vaccination during the first 2 years of life was not associated with increased rates of any of the 50 disorders assessed. For groups of combined outcomes, adjusted hazard ratios per 1-mg increase in aluminum exposure were 0.98 (95% CI, 0.94 to 1.02) for any autoimmune disorder, 0.99 (CI, 0.98 to 1.01) for any atopic or allergic disorder, and 0.93 (CI, 0.90 to 0.97) for any neurodevelopmental disorder. For most individually analyzed outcomes, the upper bounds of the 95% CIs were incompatible with relative increases greater than 10% or 30%.

Limitation: Individual medical records were not reviewed.

Conclusion: This nationwide cohort study did not find evidence supporting an increased risk for autoimmune, atopic or allergic, or neurodevelopmental disorders associated with early childhood exposure to aluminum-adsorbed vaccines. For most outcomes, the findings were inconsistent with moderate to large relative increases in risk, although small relative effects, particularly for some rarer disorders, could not be statistically excluded.

Primary Funding Source: None.

Ann Intern Med. 2025;178:000-000. doi:10.7326/ANNALS-25-00997 For author, article, and disclosure information, see end of text. This article was published at Annals.org on 15 July 2025.

A luminum-based adjuvants are commonly used in nonlive vaccines to enhance the immune response by adsorbing vaccine antigens (1, 2). Aluminumadsorbed vaccines include but are not limited to those offered in early childhood vaccination programs, such as diphtheria, tetanus, and pertussis; *Haemophilus influenzae* type b (Hib); pneumococcal conjugate vaccine (PCV); and hepatitis A and B vaccines (3).

Although immunization with aluminum-adsorbed vaccines in children has been used worldwide for many decades and is generally considered safe, concerns about potential harms continue to resurface. In particular, findings from mainly animal studies have fueled theoretical concerns about potential neurotoxic effects of aluminum-adsorbed vaccines and an increased risk for inducing autoimmunity and atopic disorders (3-10). Human data to inform these concerns are

lacking, and the available literature is mainly limited to preclinical, ecological, and smaller observational studies of selected outcomes (9-16). Large-scale population studies may be helpful in investigating such potential associations.

The Danish childhood vaccination program, established in 1943 with the introduction of diphtheria vaccination, is a nationwide initiative offering vaccines recommended by the Danish Health Authority free of

See also:

Summary for Patients Web-Only Supplement

 $^{\odot}$ 2025 American College of Physicians 1

Page: 2 Total Pages: 10 ini:

EMBARGOED FOR RELEASE UNTIL 5:00 PM ET ON MONDAY 14 JULY 2025

Original Research

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

charge to all children residing in Denmark. Vaccination coverage during the first 2 years of life is high (94% to 97% in 2023) (17). Over the past 25 years, the specific vaccines included in the program have changed over time because of policy updates, such as the introduction of pneumococcal vaccination in 2007; periodic substitutions of diphtheria, tetanus, and pertussis vaccines due to supply shortages; and the replacement of older with newer formulations. Because the vaccines have varying aluminum content, these uniformly applied policy changes have led to systematically different cumulative doses of aluminum received through childhood vaccination across birth cohorts (mainly before age 2 years). Therefore, these externally imposed systematic variations in aluminum exposure through early childhood vaccination that are independent of individual characteristics provide a natural quasiexperimental framework for estimating the effect of aluminum exposure from early childhood vaccination on health outcomes.

We leveraged Danish nationwide health care registries and these changes in vaccine use to investigate the association between cumulative aluminum exposure from vaccination with aluminum-adsorbed vaccines in the first 2 years of life and risk for chronic autoimmune, atopic or allergic, and neurodevelopmental disorders in a nationwide cohort of children born from 1 January 1997 to 31 December 2018, with follow-up through 2020.

Methods

Study Population

We established a cohort of all children born live in Denmark from 1 January 1997 to 31 December 2018 using data from the nationwide Medical Birth Registry (18). This registry contains comprehensive information on all births in Denmark, including personal identification numbers (19) for infants and parents, birth date, vital status (that is, dead or alive), maternal smoking during pregnancy, and other infant characteristics. The personal identification number enabled linkage to vaccination records, hospital-recorded diagnoses, and potential confounders at both the child and maternal level (for example, preterm birth, household income, and maternal medical history) (20).

To be included in our study, children needed to be alive at age 2 years, not have emigrated from Denmark, not have been diagnosed with certain congenital or preexisting conditions (including congenital rubella syndrome, respiratory conditions, primary immune deficiency, and heart or liver failure), and not have received an implausible number of vaccines (**Supplement Table 1**, available at Annals.org). In addition, we excluded children whose mothers had not lived in Denmark for at least 2 years before childbirth. Register-based research in Denmark is exempt from ethics committee approval under Danish law.

Vaccines and Aluminum Content

Since 1997, the Danish childhood vaccination program has offered (that is, not mandated) an aluminum-adsorbed vaccine for diphtheria, tetanus, acellular pertussis, inactivated poliovirus, and Hib (DTaP-IPV/Hib) in 3 doses in the first year of life (at 3, 5, and 12 months). Aluminum-adsorbed PCV (7-valent) was introduced in 2007 (substituted with the 13-valent vaccine in 2010) and is typically co-administered with DTaP-IPV/Hib. A DTaP-IPV booster (1997 to 2002 against diphtheria and tetanus; 2003 against diphtheria, tetanus, and pertussis; and since 2004 against diphtheria, tetanus, pertussis, and polio) is given again at age 5 years. Supplement Figure 1 (available at Annals.org) shows the chronological sequence of these national changes in the aluminum-adsorbed vaccines used before age 2 years, and Supplement
 Table 2 (available at Annals.org) provides a program
overview.

All childhood vaccinations in Denmark are administered by local general practitioners, who receive reimbursement from the Danish government on registration in the National Health Service Register (21). From this register, we obtained information on the timing, number, and types of vaccines administered to each child. The aluminum content per dose of aluminum-adsorbed vaccine ranged from 0.125 to 1.00 mg (Supplement Figure 1 and Supplement Table 2). We calculated the total aluminum exposure from all childhood vaccines received by age 2 years, which was the exposure.

Outcomes

We examined 50 adverse events, including 36 autoimmune, 9 atopic or allergic, and 5 neurodevelopmental disorders, with definitions adapted from previous work (22-28). During follow-up, we identified outcome events as any first hospital contact where an outcome diagnosis was recorded in the National Patient Registry (29) or, for some of the atopic disorders and attention deficit-hyperactivity disorder, by filling of prescriptions at any Danish pharmacy recorded in the National Prescription Registry (30) (see Supplement Table 1 for International Classification of Diseases, 10th revision, codes; Anatomical Therapeutic Chemical codes; and further details). The date of admission or filled prescription was considered the event date. We analyzed each outcome separately and included only incident events occurring after age 2 years.

Statistical Analysis

We followed children from 2 years of age for study outcome events until 31 December 2020, or until they reached age 5 years, died, or were lost to follow-up,

2 Annals of Internal Medicine • Vol. 178 No. 10 • October 2025

J_ID: AIM ART NO: ANNALS-25-00997 Date: 9-July-25

Page: 3 Total Pages: 10

EMBARGOED FOR RELEASE UNTIL

5:00 PM ET ON MONDAY 14 JULY 2025

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

whichever came first. To evaluate the potential association between cumulative aluminum exposure from vaccinations by age 2 years and the study outcomes, we used Cox proportional hazards regression. Time since age 2 years was used as the underlying time scale to estimate adjusted hazard ratios per 1-mg increase in aluminum exposure, along with 95% Cls. These hazard ratios represent how much the hazard changed when a child received 1 additional milligram of aluminum in their vaccines by age 2 years compared with a child who received 1 mg less. Each outcome model was adjusted for a priori-defined potential confounders, including birth year and season, sex, maternal age at delivery, maternal place of birth, maternal smoking during pregnancy, parity, preterm birth, birthweight, number of visits to a general practitioner before age 2 years, maternal prescription drug use during pregnancy, selected maternal conditions within 5 years before the child's birth date, and parental household income (Supplement Table 1). Outcomes with fewer than 20 cases were not analyzed separately. Missing data on smoking during pregnancy were imputed using logistic regression with all other covariates as predictors, and children with missing information on other covariates were excluded because there were few of them (Supplement Table 1). We assessed the proportional hazards assumption by visually inspecting scaled Schoenfeld residuals and testing for time scale-varying effects. When a variable violated the assumption, the baseline hazard function was stratified accordingly. Secondary analyses included stratifying by sex and birth cohort (1997 to 2006 and 2007 to 2018), removing children not vaccinated with aluminum-adsorbed vaccines by age 2 years (because these could have more different health care use), extending follow-up to age 8 years, shortening follow-up to age 3 or 4 years (to indirectly evaluate the potential for an interaction between the hazard ratio and time scale of follow-up), restricting to complete cases (to indirectly evaluate for bias introduced by our imputation method), redefining exposure and follow-up to start at age 14 months (to also capture earlier events while allowing some grace period from the 12-month appointment), and considering the total aluminum content received through early childhood vaccination as a categorical variable (separated by the median and in groups of >0 to 1.5 mg, >1.5 to 3 mg, and >3 to 4.5 mg; to examine for potential nonlinear associations not captured by the dose-response parameterization). Latter categorized exposure levels were also compared by risk differences calculated from cumulative incidences at age 5 years, which we obtained from the Kaplan-Meier estimator that we adjusted for all baseline covariates using stabilized inverse probability of treatment weights. Statistical tests were 2-sided, and analyses were done in R, ver-

sion 4.4.0 (R Foundation); see the Methods section of

the **Supplement** (available at Annals.org) for R packages and key code used for analytic modeling.

ini:

ORIGINAL RESEARCH

Role of the Funding Source

No funder had any role in the design, conduct, or reporting of the study.

RESULTS

Study Population

Figure 1 shows a flow chart of the construction of F1 the study cohort. A total of 1 224 176 children (48.8% female) born from 1 January 1997 to 31 December 2018 were included. The Table shows baseline char- T1 acteristics by the cumulative amount of aluminum exposure through vaccination. Only 15237 children (1.2%) did not receive any aluminum-adsorbed vaccines before age 2 years. Among vaccinated children, most baseline characteristics were similar, with absolute standardized mean differences below 0.1 (Supplement Figure 2, available at Annals.org); however, those who received more than 0 mg but less than 1.5 mg of aluminum had higher proportions of lower disposable household income and fewer general practitioner encounters. In addition, those who received more than 3 mg of aluminum more often had a maternal history of psychiatric disorders and diabetes. As expected, the total aluminum exposure through vaccination at age 2 years varied by birth year (Figure 2); the median was F2 3 mg (IQR, 2.8 to 3.4 mg), and the total exposure ranged from 0 to 4.5 mg. The median age at the end of follow-up was 5 years (IQR, 5 to 5 years); 6510 participants (0.5%) were censored during follow-up because of emigration or disappearance and 335 (0.03%) because of death.

Risk for Autoimmune, Atopic, and Neurodevelopmental Disorders

Figure 3 shows the number of outcome events F3 between ages 2 and 5 years within the cohort during the 24-year study period (1997 to 2020) and hazard ratios for all of the autoimmune, atopic and allergic, or neurodevelopmental disorders assessed per 1-mg increase in the cumulative aluminum exposure from vaccination during the first 2 years of life. Twenty autoimmune disorders had fewer than 20 cases and were not separately analyzed in the primary analysis (Supplement Table 3, available at Annals.org). Most autoimmune disorders were relatively rare (incidence rates ranging from 0.8 to 50.5 per 100000 personyears), with hazard ratios for specific autoimmune disorders ranging from 0.69 (95% CI, 0.47 to 1.01) to 1.13 (Cl, 0.97 to 1.31) and with a hazard ratio for the combined group of outcomes of 0.98 (Cl, 0.94 to 1.02) per 1-mg increase in the cumulative aluminum exposure from childhood vaccination. For the 16 autoimmune disorders individually studied, the upper bounds of the 95% CIs were 1.10 or lower for 8 outcomes, 1.30

Annals.org

EMBARGOED FOR RELEASE UNTIL 5:00 PM ET ON MONDAY 14 JULY 2025

Original Research

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

ini:

Figure 1. Study flow diagram.



This flow chart illustrates the construction of the study cohort, including the number of individuals excluded. Details on variable definitions are provided in Supplement Table 1 (available at Annals.org). Each outcome was studied separately for only incident events occurring after age 2 y, which is why the studied sample size varied by different outcome history exclusions.

* The total number of excluded persons by the exclusion criteria is not the sum of those meeting each exclusion criterion because some individuals were excluded for more than 1 of the stated reasons.

or lower for 5 outcomes, and between 1.31 and 1.68 for the remaining 3 outcomes.

Asthma was the most common outcome (28 346 cases), with a hazard ratio per 1-mg increase in the cumulative aluminum exposure from childhood vaccination of 0.96 (Cl, 0.94 to 0.98), followed by atopic dermatitis (22 978 cases; hazard ratio, 1.02 [Cl, 1.00 to 1.04]) and allergic rhinitis (22 841 cases; hazard ratio, 0.99 [Cl, 0.97 to 1.01]). The hazard ratio for the combined group of atopic or allergic disorder outcomes was 0.99 (Cl, 0.98 to 1.01). The upper bounds of the 95% Cls were 1.10 or lower for 8 of the 9 individually analyzed outcomes in this group and 1.79 for insect sting allergy.

The hazard ratio for the combined group of neurodevelopmental disorder outcomes was 0.93 (Cl, 0.90 to 0.97); 4806 participants had been diagnosed with autism spectrum disorder and 1580 with attention deficit-hyperactivity disorder before age 5 years. The hazard ratios per 1-mg increase in the cumulative aluminum exposure from childhood vaccination before age 2 years were 0.93 (Cl, 0.89 to 0.97) and 0.90 (Cl, 0.84 to 0.96), respectively. In this primary analysis, except for Asperger syndrome (hazard ratio, 1.13 [Cl, 0.89 to 1.44]) and atypical autism (hazard ratio, 0.94 [Cl, 0.79 to 1.12]), estimates for the individual outcomes were incompatible with any increased risk, with the upper bounds of the 95% Cls below 1.00.

The hazard ratios and corresponding 95% Cls were qualitatively similar in the secondary analyses when we stratified by sex (Supplement Figure 3, available at Annals.org) or birth year (Supplement Figure 4, available at Annals.org), excluded unvaccinated children (Supplement Figure 5, available at Annals.org), redefined exposure and started follow-up at age 14 months instead of 2 years (Supplement Figure 6, available at Annals.org), ended follow-up at age 3 or 4 years (Supplement Figure 7, available at Annals. org), or restricted to complete cases (Supplement Figure 8, available at Annals.org). Categorizing cumulative aluminum exposure from childhood vaccination before age 2 years by the median (Supplement Figure 9, available at Annals.org) and levels received (>0 to 1.5 mg, >1.5 to 3 mg, or >3 to 4.5 mg) (Supplement Figures 10 and 11, available at Annals.org) did not suggest nonlinear associations. Extending follow-up to age 8 years allowed more outcome events to be analyzed separately (specifically, 20 of all 36 included autoimmune outcomes had ≥20 cases in this analysis) and yielded similar results to our main findings (Supplement Figure 12, available at Annals.org). For example, by age 8 years, 12 126 children were diagnosed with autism spectrum disorder (hazard ratio, 0.95 [Cl, 0.92 to 0.97] per 1-mg increase in the cumulative aluminum exposure from childhood vaccination) and 12091 with attention deficithyperactivity disorder (hazard ratio, 0.92 [Cl, 0.90 to 0.94]). Except for Asperger syndrome (hazard ratio, 1.02 [Cl, 0.93 to 1.12]) and atypical autism (hazard ratio, 0.95

4 Annals of Internal Medicine • Vol. 178 No. 10 • October 2025

Total Pages: 10 ini:

Page: 5

EMBARGOED FOR RELEASE UNTIL

5:00 PM ET ON MONDAY 14 JULY 2025

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

ORIGINAL RESEARCH

Table. Baseline Characteristics of a Nationwide Cohort of Children Born From 1997 to 2018 in Denmark, by Cumulative Aluminum Exposure Through Early Childhood Vaccination*

Characteristic	Cumulative	No Receipt of Aluminum-Adsorbed		
	>0-1.5 mg (<i>n</i> = 42 990)	>1.5-3 mg (<i>n</i> = 701 571)	>3-4.5 mg (n = 464 378)	Vaccines (<i>n</i> = 15237)
Female sex	20 826 (48.4)	342 172 (48.8)	227 381 (49.0)	7423 (48.7)
Preterm birth	3327 (7.7)	49 527 (7.1)	31 085 (6.7)	1099 (7.2)
Birthweight				
<1500 g	364 (0.8)	4587 (0.7)	3086 (0.7)	147 (1.0)
1500-<2500 g	1924 (4.5)	28 897 (4.1)	19339 (4.2)	610 (4.0)
≥2500 g	40 702 (94.7)	668087 (95.2)	441 953 (95.2)	14 480 (95.0)
Year of birth				
1997-2001	14 697 (34.2)	268 552 (38.3)	0 (0)	5655 (37.1)
2002-2006	10 842 (25.2)	249 620 (35.6)	26 055 (5.6)	3481 (22.8)
2007-2011	6088 (14.2)	36 096 (5.1)	235 177 (50.6)	2264 (14.9)
2012-2016	7319 (17.0)	90 935 (13.0)	155 483 (33.5)	2517 (16.5)
2017-2018	4044 (9.4)	56 368 (8.0)	47 663 (10.3)	1320 (8.7)
e (1).1				
Season of birth	10 402 (24 4)	100.040 (25.0)	110 400 (22.0)	2022 (25.2)
Spring	10483 (24.4)	180 842 (25.8)	110 499 (23.8) 125 635 (27.1)	3833 (25.2)
Summer Autumn	10 509 (24.4) 10 764 (25.0)	187 602 (26.7) 172 008 (24.5)	119 472 (25.7)	3920 (25.7)
Winter	11 234 (26.1)	161 119 (23.0)	108 772 (23.4)	3789 (24.9) 3695 (24.3)
Winter	11234 (20.1)	101117 (23.0)	100772 (23.4)	5075 (24.5)
Number of contacts with a general practitioner				
Tertile 1 (lowest)	25859(60.2)	235 936 (33.6)	147 813 (31.8)	12 452 (81.7)
Tertile 2	9419 (21.9)	219 163 (31.2)	178 316 (38.4)	1539 (10.1)
Tertile 3 (highest)	7712 (17.9)	246 472 (35.1)	138 249 (29.8)	1246 (8.2)
Maternal age at delivery				
<20 y	805 (1.9)	9145 (1.3)	5231 (1.1)	233 (1.5)
20-24 y	5390 (12.5)	79686(11.4)	46 893 (10.1)	1770 (11.6)
25-29 y	14060 (32.7)	243 872 (34.8)	144 807 (31.2)	4550 (29.9)
30-34 y	14660 (34.1)	248 217 (35.4)	170 392 (36.7)	5089 (33.4)
35-39 y	6764 (15.7)	102 902 (14.7)	81 109 (17.5)	2877 (18.9)
≥40 y	1311 (3.0)	17 749 (2.5)	15 946 (3.4)	718 (4.7)
Maternal place of birth				
Denmark	36 640 (85.2)	616251 (87.8)	401 666 (86.5)	12 484 (81.9)
Europe	2676 (6.2)	36 431 (5.2)	29 946 (6.4)	1393 (9.1)
Outside Europe	3674 (8.5)	48 889 (7.0)	32 766 (7.1)	1360 (8.9)
Disposable household income				
Quantile 1 (lowest)	13295 (30.9)	167 683 (23.9)	92 405 (19.9)	5836 (38.3)
Quantile 2	11 302 (26.3)	199 959 (28.5)	95 573 (20.6)	3949 (25.9)
Quantile 3	9623 (22.4)	181 060 (25.8)	123 538 (26.6)	3007 (19.7)
Quantile 4 (highest)	8770 (20.4)	152 869 (21.8)	152 862 (32.9)	2445 (16.0)
Maternal parity	1 (500 (00 1)		04/470/4/ ()	5000 (05.4)
1	16 509 (38.4)	304 972 (43.5)	216 179 (46.6)	5392 (35.4)
2	16119 (37.5)	264 238 (37.7)	170 512 (36.7)	5536 (36.3)
≥3	10362 (24.1)	132361 (18.9)	77 687 (16.7)	4309 (28.3)
Maternal smoking during pregnancy	8221 (19.1)	112807(16.1)	53 621 (11.5)	2545 (16.7)
Maternal medical history during the past 5 y				
Atopic disorder	6571 (15.3)	116 629 (16.6)	65 389 (14.1)	1973 (12.9)
Neurodevelopmental disorder	152 (0.4)	1495 (0.2)	2031 (0.4)	51 (0.3)
Autoimmune disorder	1183 (2.8)	18 849 (2.7)	18 923 (4.1)	454 (3.0)
Psychiatric disorder Diabetes	2101 (4.9) 1387 (3.2)	25 491 (3.6) 19 892 (2.8)	28 591 (6.2) 22 368 (4.8)	594 (3.9) 391 (2.6)
		1/0/2 (2.0)	22 300 (4.0)	571(2.0)
Maternal prescription drug use during pregnance Antibiotics	y 14687 (34.2)	223 078 (31.8)	165 210 (25 6)	4452 (29.2)
Systemic corticosteroids	241 (0.6)	3775 (0.5)	165210(35.6) 3721(0.8)	74 (0.5)
Immunosuppressants	34 (0.1)	441 (0.1)	606 (0.1)	≤3 (0.0)
Opioids	511 (1.2)	6198 (0.9)	4907 (1.1)	151 (1.0)
Antiepileptics	204 (0.5)	3328 (0.5)	2655 (0.6)	65 (0.4)

* Values are numbers (percentages). Cells with ≤3 participants (but not 0) cannot be reported to ensure privacy preservation. Details on variable definitions are provided in Supplement Table 1 (available at Annals.org).

Page: 6 Total Pages: 10 ini:

EMBARGOED FOR RELEASE UNTIL

Original Research

5:00 PM ET ON MONDAY 14 JULY 2025 Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

Figure 2. Early childhood vaccination with aluminum-adsorbed vaccines in Denmark from 1997 through 2020.



The figure shows the distribution of cumulative aluminum exposure (in milligrams) through vaccination at age 2 y across the study cohort grouped by birth year. The national changes in the aluminum-adsorbed vaccines used in the Danish childhood vaccination program during the study period and their respective aluminum content are shown in Supplement Figure 1 (available at Annals.org).

[CI, 0.88 to 1.03]), estimates for the individual neurodevelopmental outcomes assessed were incompatible with any increases in risk, with the upper bounds of the 95% Cls equal to or below 1.00. With follow-up until age 8 years, the upper bounds were below 1.10 for 17 of 29 individual autoimmune or atopic or allergic outcomes, below 1.30 for 7 outcomes, and between 1.33 and 2.81 for the remaining 5.

DISCUSSION

In this nationwide cohort study of approximately 1.2 million children, estimates of the associations between cumulative aluminum exposure from early childhood vaccination and chronic autoimmune, atopic or allergic, and neurodevelopmental disorders were incompatible with moderate to large relative increases in risk and most consistent with no increased risk, but small relative increases in risk could not be statistically excluded, particularly for some rarer outcomes.

This work was motivated by ongoing concerns about potential harms of aluminum-adsorbed nonlive vaccines. A recent cohort study by Daley and colleagues (16) using data from the U.S. Vaccine Safety Datalink examined the risk for asthma in relation to cumulative aluminum exposure from childhood vaccination in 326 991 children. They identified 7546 children with asthma by age 5 years (2.3%) and reported estimates compatible with small to moderate relative increases in risk per 1-mg aluminum exposure through vaccination before age 2 years: adjusted hazard ratios of 1.26 (Cl, 1.07 to 1.49) and 1.19 (Cl, 1.14 to 1.25) among children with and without eczema, respectively (16). As the authors also noted, important potential

confounders were unavailable, such as maternal smoking, atopic predisposition, and socioeconomic status. In our study, analyzing up to 38045 asthma cases (5.0% of the studied population for this outcome), the estimated hazard ratio per 1-mg increase in cumulative aluminum exposure from early childhood vaccination was 0.96 (CI, 0.94 to 0.98), with the upper confidence limit incompatible with even small relative increases in risk. Vaccination uptake was lower in the study by Daley and colleagues: 60.6% of participants were fully vaccinated, and effect sizes were attenuated when analyses were restricted to this subpopulation, which could suggest residual confounding (16). We found similar estimates for other atopic disorders and a hazard ratio of 0.99 (CI, 0.98 to 1.01) for the combined group of atopic or allergic disorders.

Neurotoxic effects of aluminum have been observed in animal models (rats) subjected to long-term, highdose parenteral administration and in humans receiving high-level exposure through dialysis, occupational inhalation, or intravenous parenteral nutrition (6, 31-33). These findings, along with ecological studies, have contributed to the ongoing concerns about aluminum exposure from childhood vaccinations potentially contributing to the development of neurodevelopmental disorders, such as autism spectrum disorder and attention deficit-hyperactivity disorder, as well as autoimmunity (6, 9, 14). However, a previous study suggests that the aluminum levels in infants from vaccinations are well below established minimal risk level thresholds (11). In a cross-sectional study of 85 infants, aluminum concentrations in blood (Spearman $\rho = -0.13$; P = 0.26) and hair (Spearman $\rho = 0.06$; P = 0.56) did not seem to be positively correlated with receipt of aluminum-adsorbed vaccines, but the precision of these estimates was not reported (12). Similarly, a small study of 15 preterm infants reported no measurable change in serum aluminum concentrations after vaccination with aluminumadsorbed vaccines, although no specific statistics were provided (13). Despite these preliminary findings, largescale studies directly evaluating these potential associations between aluminum exposure from vaccines and chronic diseases in childhood have been lacking.

Our study, based on a nationwide cohort of about 1.2 million children with analyses across a 24-year period, offers a large-scale, population-based evaluation of the safety of aluminum-adsorbed vaccines in childhood vaccination programs. The upper bounds of the 95% CIs for the combined outcome groups were incompatible with relative risk increases greater than 2% for autoimmune disorders, 1% for atopic or allergic disorders, and any increase for neurodevelopmental disorders, per 1-mg increase in aluminum exposure through early childhood vaccination. In addition, in the primary analysis of the 30 individually analyzed outcomes, the upper bounds ruled out relative increases greater than 10% for 19 outcomes,

Total Pages: 10 ini:

Page: 7

EMBARGOED FOR RELEASE UNTIL

5:00 PM ET ON MONDAY 14 JULY 2025

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

ORIGINAL RESEARCH

Figure 3. Association between cumulative aluminum exposure through early childhood vaccination and chronic disease in children.

Outcome	Cases, n	Person-Years		Adjusted Hazard Ratio (95% CI) per 1-mg Increase in Aluminum Received Through Vaccines by Age 2 y
Autoimmune outcomes	5174	3 400 369	*	0.98 (0.94–1.02)
Rheumatoid arthritis	51	3 415 683		1.10 (0.72–1.68)
Hashimoto thyroiditis	26	3 415 737	_	0.92 (0.54-1.56)
Idiopathic thrombocytopenic purpura	422	3 414 006	·	1.13 (0.97–1.31)
Hypothyroidism	104	3 414 265		0.98 (0.74-1.30)
Vitiligo	39	3 415 724		0.80 (0.53-1.23)
Kawasaki disease	266	3 414 864	·	1.00 (0.83-1.19)
Juvenile arthritis	809	3 413 561		1.05 (0.94–1.16)
Type 1 diabetes	669	3 414 203		1.03 (0.92–1.15)
Vasculitis unspecified	97	3 415 308		0.84 (0.64–1.10)
Celiac disease	652	3 414 085		0.98 (0.87–1.10)
Crohn disease	85	3 415 496		0.82 (0.62–1.07)
Psoriasis	114	3 415 456		0.81 (0.64–1.03)
Henoch-Schönlein purpura	1722	3 412 356	-	0.96 (0.90–1.03)
Myositis	40	3 415 678	•	0.69 (0.47–1.01)
Ulcerative colitis	68	3 415 469		0.72 (0.52–0.98)
Erythema nodosum	112	3 415 477		0.74 (0.58–0.95)
Atopic and allergic outcomes	39 465	1 562 490	•	0.99 (0.98–1.01)
Insect sting allergy	33	3 415 745		- 1.09 (0.66–1.79)
Atopic dermatitis	22 978	2 703 626	÷	1.02 (1.00-1.04)
Drug allergy	2275	3 405 358	- •	0.96 (0.90-1.02)
Allergic rhinoconjunctivitis	22 841	3 174 495		0.99 (0.97-1.01)
Asthma	28346	2 032 343	▲	0.96 (0.94-0.98)
Angioedema and urticaria	3435	3 399 403	▲	0.90 (0.86-0.95)
Allergy unspecified	1621	3 409 018	.	0.86 (0.80–0.92)
Anaphylaxis or epinephrine autoinjector	1372	3 411 576		0.84 (0.78–0.91)
Food allergy	2288	3 399 832	.	0.84 (0.79–0.89)
Neurodevelopmental outcomes	5200	3 409 893	•	0.93 (0.90–0.97)
Asperger syndrome	175	3 415 664	· · · ·	1.13 (0.89–1.44)
Atypical autism	276	3 415 524		0.94 (0.79-1.12)
Autistic disorder	2686	3 412 494	-	0.94 (0.89-0.99)
Autism spectrum disorder composite	4806	3 414 259	•	0.93 (0.89-0.97)
Attention deficit-hyperactivity disorder	1580	3 414 493	-	0.90 (0.84-0.96)
Other pervasive developmental disorders	1702	3 413 647	- + -	0.89 (0.83–0.95)
			0.5 0.7 1 1.4	-
			Hazard Ratio (Log Scale)	

The figure reports adjusted hazard ratios for the association between the cumulative aluminum exposure from vaccination with aluminum-adsorbed vaccines by age 2 y and risk for autoimmune, atopic or allergic, and neurodevelopmental disorders diagnosed between ages 2 and 5 y in children born in Denmark between 1997 and 2018, with follow-up through 2020 (ordered [descending] by the upper bound of the 95% CI). Each outcome was studied separately, which is why there may be differences in the denominators due to different exclusions. Children were followed from age 2 y until age 5 y for respective outcome events. Hazard ratios were adjusted for sex, calendar year and season of birth, preterm birth, birthweight, number of visits to the general practitioner before age 2 y, maternal age at delivery, maternal place of birth, maternal smoking during pregnancy, maternal health conditions, maternal prescription drug use during pregnancy, and parental household income. Outcomes with fewer than 20 cases were not separately analyzed (Supplement Table 3, available at Annals.org).

greater than 30% for 7 outcomes, and between 31% and 79% for the 4 remaining rarer outcomes (the latter 4 outcomes had incidence rates ranging from 7.6 to 14.9 per 1 million person-years).

Our study has limitations. First, because we analyzed data from routine clinical practice, our study lacked true randomization of exposure. Instead, we leveraged the variation in aluminum content of vaccines administered in early childhood over the 24-year study period resulting from systematic national changes in the Danish childhood vaccination program. In addition, we adjusted our analyses for a wide range of potential confounders, including socioeconomic characteristics. Nonetheless, residual confounding cannot be excluded, particularly if unmeasured confounding factors were not adequately indirectly accounted for through the included set of covariates. Second, as expected, children born later in the study period were on average exposed to more aluminum through early childhood vaccination than those born earlier. The prevalence of diagnoses for several analyzed chronic disorders has increased over the past decades (34-38). If no true association exists, these parallel trends in data would tend to skew results toward positive associations, unlike what we observed. Stratifying by birth cohort also did not change the study findings. Third, very early prediagnosis symptoms could potentially either reduce or increase the propensity

Page: 8 Total Pages: 10 ini:

EMBARGOED FOR RELEASE UNTIL 5:00 PM ET ON MONDAY 14 JULY 2025 Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

Original Research

could not be statistically excluded, particularly for

some rarer outcomes.

Denmark (A.H.).

to vaccination, which could bias toward underestimating or overestimating, respectively, the risk in vaccinated children. However, as mentioned above, our analyses are primarily based on the random variation in aluminum exposure through vaccination, in contrast to a vaccinated versus unvaccinated comparison. Fourth, to ascertain our exposure of aluminum-adsorbed vaccination, we used the Danish National Health Service Register wherein registration is required by the general practitioner to receive reimbursement from the Danish government. This implies a high degree of validity. Any exposure misclassification is less likely associated with our outcomes and would bias results toward the null. Of note, although our primary continuous exposure definition most likely mitigates potential biases related to temporal outcome trends, exposure misclassification, and unmeasured confounding from differences in health care use and behaviors, the secondary analyses using categorical exposure comparisons are more susceptible to such biases. Fifth, we studied incident cases of 50 different chronic disorders, some of which were very rare and some of which typically develop or are diagnosed after age 5 years. Consequently, despite our nationwide longitudinal data permitting a large sample size, some outcomes could not be individually analyzed. Cases were defined through hospital diagnoses or as prescription records for disorder-related medications (for atopic or attention deficit-hyperactivity disorder). Of note, the validity of each individual outcome has not been systematically assessed; however, previous evaluations suggest that the National Patient Registry has sufficient data quality overall for recorded diagnoses, including several of the outcomes studied here (or closely defined proxies) (39). Outcome misclassification would tend to be less likely for more severe or rare disorders and is unlikely to be influenced by the extent of aluminum exposure through vaccination in early childhood. However, potential uncaptured disorders in early years (such as through underdiagnosis) relative to later years would tend to skew our estimates toward increased risks. Although associations with the studied disorders when emerging later in life cannot be ruled out, extending follow-up to age 8 years did not indicate such trends. Our results may be generalizable to similar pediatric populations, although their applicability to adult settings is more uncertain. In addition, our findings may not directly generalize to clinical scenarios not studied, such as the use of other vaccines in early childhood (for example, with cumulative aluminum amounts received outside our cohort exposure distribution) or risk for other chronic disorders.

In conclusion, in this nationwide cohort study of approximately 1.2 million children, findings were incompatible with moderate to large relative increases in the risks for autoimmune, atopic or allergic, and neurodevelopmental disorders associated with early childhood exposure to aluminum-adsorbed vaccines for most outcomes, although small relative increases From Department of Epidemiology Research, Statens Serum Institut, Copenhagen, Denmark (N.W.A., I.B.S.); National Allergy Research Centre, Department of Dermatology and Allergy, Herlev and Gentofte Hospital, University of Copenhagen, Hellerup, Denmark (S.S.H.); and Department of Epidemiology Research, Statens Serum Institut, and Pharmacovigilance Research Center, Department of Drug Design and Pharmacology, Faculty of Health and Medical Sciences, University of Copenhagen, Copenhagen,

Note: Dr. Andersson and Ms. Bech Svalgaard had full access to all of the data in the study and take responsibility for the integrity of the data and the accuracy of the data analyses.

Disclosures: Disclosure forms are available with the article online.

Reproducible Research Statement: *Study protocol:* Not available. *Statistical code:* See the Methods section of the **Supplement** (available at Annals.org). *Data set:* No additional data are available. Owing to data privacy regulations in Denmark, the raw data cannot be shared. However, the data are available for research on reasonable request to the Danish Health Data Authority and within the framework of the Danish data protection legislation and any required permission from authorities.

Corresponding Author: Niklas Worm Andersson, MD, PhD, Department of Epidemiology Research, Statens Serum Institut, Artillerivej 5, 2300 Copenhagen S, Denmark; e-mail, nian@ssi. dk.

Author contributions are available at Annals.org.

References

1. Clements CJ, Griffiths E. The global impact of vaccines containing aluminium adjuvants. Vaccine. 2002;20 Suppl 3:S24-S33. [PMID: 12184361] doi:10.1016/s0264-410x(02)00168-8

2. Marrack P, McKee AS, Munks MW. Towards an understanding of the adjuvant action of aluminium. Nat Rev Immunol. 2009;9:287-293. [PMID: 19247370] doi:10.1038/nri2510

3. Hogenesch H. Mechanism of immunopotentiation and safety of aluminum adjuvants. Front Immunol. 2012;3:406. [PMID: 23335921] doi:10.3389/fimmu.2012.00406

4. Tomljenovic L, Shaw CA. Do aluminum vaccine adjuvants contribute to the rising prevalence of autism? J Inorg Biochem. 2011;105:1489-1499. [PMID: 22099159] doi:10.1016/j.jinorgbio.2011.08.008

5. Tomljenovic L, Shaw CA. Aluminum vaccine adjuvants: are they safe? Curr Med Chem. 2011;18:2630-2637. [PMID: 21568886] doi:10.2174/092986711795933740

6. Willhite CC, Karyakina NA, Yokel RA, et al. Systematic review of potential health risks posed by pharmaceutical, occupational and consumer exposures to metallic and nanoscale aluminum, aluminum oxides, aluminum hydroxide and its soluble salts. Crit Rev Toxicol. 2014;44 Suppl 4:1-80. [PMID: 25233067] doi:10.3109/ 10408444.2014.934439

7. Gherardi RK, Crépeaux G, Authier FJ. Myalgia and chronic fatigue syndrome following immunization: macrophagic myofasciitis

Page: 9 Total Pages: 10

EMBARGOED FOR RELEASE UNTIL 5:00 PM ET ON MONDAY 14 JULY 2025

Aluminum-Adsorbed Vaccines and Chronic Diseases in Childhood

and animal studies support linkage to aluminum adjuvant persistency and diffusion in the immune system. Autoimmun Rev. 2019;18:691-705. [PMID: 31059838] doi:10.1016/j.autrev.2019.05.006

8. Sastry M, Zhang B, Chen M, et al. Adjuvants and the vaccine response to the DS-Cav1-stabilized fusion glycoprotein of respiratory syncytial virus. PLoS One. 2017;12:e0186854. [PMID: 29073183] doi:10.1371/journal.pone.0186854

9. Conklin L, Hviid A, Orenstein WA, et al. Vaccine safety issues at the turn of the 21st century. BMJ Glob Health. 2021;6:e004898. [PMID: 34011504] doi:10.1136/bmjgh-2020-004898

10. DeStefano F, Bodenstab HM, Offit PA. Principal controversies in vaccine safety in the United States. Clin Infect Dis. 2019;69:726-731. [PMID: 30753348] doi:10.1093/cid/ciz135

11. Mitkus RJ, King DB, Hess MA, et al. Updated aluminum pharmacokinetics following infant exposures through diet and vaccination. Vaccine. 2011;29:9538-9543. [PMID: 22001122] doi:10.1016/ j.vaccine.2011.09.124

12. Karwowski MP, Stamoulis C, Wenren LM, et al. Blood and hair aluminum levels, vaccine history, and early infant development: a crosssectional study. Acad Pediatr. 2018;18:161-165. [PMID: 28919482] doi:10.1016/j.acap.2017.09.003

13. Movsas TZ, Paneth N, Rumbeiha W, et al. Effect of routine vaccination on aluminum and essential element levels in preterm infants. JAMA Pediatr. 2013;167:870-872. [PMID: 23856981] doi:10.1001/ jamapediatrics.2013.108

14. Boretti A. Reviewing the association between aluminum adjuvants in the vaccines and autism spectrum disorder. J Trace Elem Med Biol. 2021;66:126764. [PMID: 33930617] doi:10.1016/j.jtemb. 2021.126764

15. Principi N, Esposito S. Aluminum in vaccines: does it create a safety problem? Vaccine. 2018;36:5825-5831. [PMID: 30139653] doi:10.1016/j.vaccine.2018.08.036

16. Daley MF, Reifler LM, Glanz JM, et al. Association between aluminum exposure from vaccines before age 24 months and persistent asthma at age 24 to 59 months. Acad Pediatr. 2023;23:37-46. [PMID: 36180331] doi:10.1016/j.acap.2022.08.006

17. Danish Health Authority. The Danish childhood vaccination program. Accessed at www.sst.dk/da/Fagperson/Graviditet-ogsmaaboern/Barnets-sundhed/Vaccination-af-boern/Boernevacci nationsprogrammet on 8 December 2024.

18. Knudsen LB, Olsen J. The Danish Medical Birth Registry. Dan Med Bull. 1998;45:320-323. [PMID: 9675544]

19. Pedersen CB. The Danish Civil Registration System. Scand J Public Health. 2011;39:22-25. [PMID: 21775345] doi:10.1177/ 1403494810387965

20. Schmidt M, Schmidt SAJ, Adelborg K, et al. The Danish health care system and epidemiological research: from health care contacts to database records. Clin Epidemiol. 2019;11:563-591. [PMID: 31372058] doi:10.2147/CLEP.S179083

21. Andersen JS, Olivarius NDF, Krasnik A. The Danish National Health Service Register. Scand J Public Health. 2011;39:34-37. [PMID: 21775348] doi:10.1177/1403494810394718

22. Arnheim-Dahlström L, Pasternak B, Svanström H, et al. Autoimmune, neurological, and venous thromboembolic adverse events after immunisation of adolescent girls with quadrivalent human papillomavirus vaccine in Denmark and Sweden: cohort study. BMJ. 2013;347:f5906. [PMID: 24108159] doi:10.1136/bmj. f5906

23. Frisch M, Besson A, Clemmensen KKB, et al. Quadrivalent human papillomavirus vaccination in boys and risk of autoimmune diseases, neurological diseases and venous thromboembolism. Int J Epidemiol. 2018;47:634-641. [PMID: 29425361] doi:10.1093/ije/ dyx273

24. Rubak S, Høst A, Christensen LB, et al. Validity of asthma diagnoses and patterns of anti-asthmatic drug use in a cohort of 2053 Danish children. Health Sci Rep. 2018;1:e77. [PMID: 30623100] doi:10.1002/hsr2.77

ini:

25. Henriksen L, Simonsen J, Haerskjold A, et al. Incidence rates of atopic dermatitis, asthma, and allergic rhinoconjunctivitis in Danish and Swedish children. J Allergy Clin Immunol. 2015;136:360-366. e2. [PMID: 25828267] doi:10.1016/j.jaci.2015.02.003

26. Noh Y, Jeong HE, Choi A, et al. Prenatal and infant exposure to acid-suppressive medications and risk of allergic diseases in children. JAMA Pediatr. 2023;177:267-277. [PMID: 36622684] doi:10. 1001/jamapediatrics.2022.5193

27. Mikkelsen AP, Greiber IK, Scheller NM, et al. Association of labor epidural analgesia with autism spectrum disorder in children. JAMA. 2021;326:1170-1177. [PMID: 34581738] doi:10.1001/jama. 2021.12655

28. Hviid A, Melbye M, Pasternak B. Use of selective serotonin reuptake inhibitors during pregnancy and risk of autism. N Engl J Med. 2013;369:2406-2415. [PMID: 24350950] doi:10.1056/ NEJMoa1301449

29. Lynge E, Sandegaard JL, Rebolj M. The Danish National Patient Register. Scand J Public Health. 2011;39:30-33. [PMID: 21775347] doi:10.1177/1403494811401482

30. Wallach Kildemoes H, Toft Sørensen H, Hallas J. The Danish National Prescription Registry. Scand J Public Health. 2011;39:38-41. [PMID: 21775349] doi:10.1177/1403494810394717

31. Lu ZY, Gong H, Amemiya T. Aluminum chloride induces retinal changes in the rat. Toxicol Sci. 2002;66:253-260. [PMID: 11896292] doi:10.1093/toxsci/66.2.253

32. Miu AC, Andreescu CE, Vasiu R, et al. A behavioral and histological study of the effects of long-term exposure of adult rats to aluminum. Int J Neurosci. 2003;113:1197-1211. [PMID: 12959739] doi:10.1080/00207450390232292

33. Bishop NJ, Morley R, Day JP, et al. Aluminum neurotoxicity in preterm infants receiving intravenous-feeding solutions. N Engl J Med. 1997;336:1557-1561. [PMID: 9164811] doi:10. 1056/NEJM199705293362203

34. Polanczyk GV, Willcutt EG, Salum GA, et al. ADHD prevalence estimates across three decades: an updated systematic review and meta-regression analysis. Int J Epidemiol. 2014;43:434-442. [PMID: 24464188] doi:10.1093/ije/dyt261

35. Zeidan J, Fombonne E, Scorah J, et al. Global prevalence of autism: a systematic review update. Autism Res. 2022;15:778-790. [PMID: 35238171] doi:10.1002/aur.2696

36. Warren CM, Sehgal S, Sicherer SH, et al. Epidemiology and the growing epidemic of food allergy in children and adults across the globe. Curr Allergy Asthma Rep. 2024;24:95-106. [PMID: 38214821] doi:10.1007/s11882-023-01120-y

37. Conrad N, Misra S, Verbakel JY, et al. Incidence, prevalence, and co-occurrence of autoimmune disorders over time and by age, sex, and socioeconomic status: a population-based cohort study of 22 million individuals in the UK. Lancet. 2023;401:1878-1890. [PMID: 37156255] doi:10.1016/S0140-6736(23)00457-9

38. Strachan DP, Rutter CE, Asher MI, et al; Global Asthma Network Phase I Study Group. Worldwide time trends in prevalence of symptoms of rhinoconjunctivitis in children: Global Asthma Network Phase I. Pediatr Allergy Immunol. 2022;33:e13656. [PMID: 34453861] doi:10.1111/pai.13656

39. Schmidt M, Schmidt SAJ, Sandegaard JL, et al. The Danish National Patient Registry: a review of content, data quality, and research potential. Clin Epidemiol. 2015;7:449-490. [PMID: 26604824]

ORIGINAL RESEARCH

EMBARGOED FOR RELEASE UNTIL 5:00 PM ET ON MONDAY 14 JULY 2025

Author Contributions: Conception and design: N.W. Andersson,

I. Bech Svalgaard, A. Hviid.

Analysis and interpretation of the data: N.W. Andersson, I. Bech Svalgaard, A. Hviid.

Drafting of the article: N.W. Andersson.

Critical revision for important intellectual content: N.W. Andersson, I. Bech Svalgaard, S.S. Hoffmann, A. Hviid.

Final approval of the article: N.W. Andersson, I. Bech Svalgaard, S.S. Hoffmann, A. Hviid.

Statistical expertise: I. Bech Svalgaard.

Collection and assembly of data: I. Bech Svalgaard, S.S. Hoffmann.